Author's response to reviews

Title: Leiomyomatosis peritonealis disseminata in association with Currarino syndrome?

Authors:

Carmine Nappi (nappi@unina.it)
Attilio Di Spiezio Sardo (cdispie@tin.it)
Vincenzo Dario Mandato (dariomandato@virgilio.it)
Giuseppe Bifulco (giuseppebifulco@hotmail.com)
Elisa Merello (m73eli@libero.it)
Antonio Savanelli (savanelli@unina.it)
Chiara Mignogna (chiaramignogna@tin.it)
Valeria Capra (valeriacapra@ospedale-gaslini.ge.it)
Maurizio Guida (maurizio.guida@unina.it)

Version: 7 Date: 27 April 2006

Author's response to reviews:

We removed 'We report such a case:' from the beginning of the Case presentation section of the abstract so that this section starts by 'A 27 year-old patient referred...' The text has been modified according to the referee's suggestion.